Contents lists available at ScienceDirect



Science of the Total Environment

journal homepage: www.elsevier.com/locate/scitotenv



Research Paper Digital twins for dynamic life cycle assessment in the built environment



Ioan Petri[®], Amin Amin[®]*, Ali Ghoroghi[®], Andrei Hodorog[®], Yacine Rezgui

School of Engineering, Cardiff University, Cardiff, CF24 3AA, United Kingdom

ARTICLE INFO

Editor: Jacopo Bacenetti

Keywords: Life cycle assessment Digital twin Environmental impact Buildings Sustainability

ABSTRACT

Dynamic life cycle assessment (LCA) integrated with digital twin technologies is emerging as a transformative approach to evaluating and managing environmental performance in the built environment. This study presents the Building Life-cycle Digital Twin (BLDT) framework—a novel methodology that combines real-time data from Internet of Things (IoT) devices, machine learning algorithms, and semantic interoperability to deliver dynamic, predictive, and high-resolution LCA for construction and infrastructure systems.

The framework, developed within the Computational Urban Sustainability Platform (CUSP), addresses the limitations of traditional static LCA by enabling continuous, data-driven sustainability assessments. Incorporating predictive modelling, BLDT empowers stakeholders with timely insights into energy use, emissions, and health and safety performance, supporting proactive environmental decision-making.

Validated through a case study at the Port of Grimsby, the BLDT framework facilitated a 25% reduction in energy consumption while enhancing operational efficiency. These results demonstrate the model's potential to support decarbonisation strategies, regulatory compliance, and long-term planning in the construction sector. By operationalising dynamic LCA through digital twins, this research contributes to the advancement of real-time sustainability analytics and resilient urban development.

1. Introduction

Digital twins are increasingly employed within the built environment to facilitate real-time monitoring, performance optimisation, and environmental assessment across the life cycle of buildings and infrastructure. Their integration into Life Cycle Assessment (LCA) represents a significant advancement, offering the capacity to assess environmental impacts dynamically rather than retrospectively. Traditional LCA methodologies, while robust for static evaluations, often lack the responsiveness to temporal and spatial variability, system dynamics, and evolving environmental conditions (Zheng et al., 2017). This inherent rigidity limits their applicability in data-rich, rapidly changing contexts such as smart construction. Accordingly, there is a growing imperative for dynamic LCA approaches capable of incorporating real-time data and localised factors, thereby yielding more accurate, actionable, and context-specific insights (Fnais et al., 2022; Ghoroghi et al., 2024).

Dynamic digital twins – virtual counterparts of physical systems that continuously synchronise with real-world operations – offer a promising avenue for overcoming these limitations (Madni et al., 2019). Unlike conventional digital representations, dynamic twins are characterised by continuous data integration, predictive modelling, and feedback capabilities. In construction and the broader built environment, they enable LCA that reflect real operational conditions rather than hypothetical scenarios. Demonstrated successes in industries such as aerospace and manufacturing – where digital twins improve maintenance planning, cost control, and operational efficiency – underscore their potential in construction applications (Xie et al., 2019; Deng et al., 2021). Within building systems, digital twins can optimise construction processes, monitor operational performance, and support end-of-life planning, aligning closely with sustainable development goals (Akinshipe et al., 2022; Borjigin et al., 2022).

The convergence of emerging technologies – digital twins, the Internet of Things (IoT), and blockchain – is reshaping environmental assessment in the built environment. Digital twins have shown potential to enhance energy efficiency and reduce carbon emissions via real-time performance optimisation (Kumar et al., 2019). IoT devices support this by continuously capturing detailed environmental data, such as energy use and emissions (Chen et al., 2022), thereby improving the granularity and relevance of LCA metrics. Blockchain technology adds transparency and trust, offering a decentralised, secure platform for data sharing among stakeholders in complex civil projects (Höjer and Wangel, 2015). Together, these technologies enable precise tracking of embodied and operational carbon, allowing for responsive and targeted emission reduction strategies throughout a building's life cycle (Mukherjee et al., 2022).

* Corresponding author.

https://doi.org/10.1016/j.scitotenv.2025.179930

Received 17 February 2025; Received in revised form 10 June 2025; Accepted 14 June 2025 Available online 3 July 2025

0048-9697/© 2025 The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

E-mail addresses: petrii@cardiff.ac.uk (I. Petri), AminA8@cardiff.ac.uk (A. Amin), ghoroghi@cardiff.ac.uk (A. Ghoroghi), hodoroga@cardiff.ac.uk

⁽A. Hodorog), rezguiy@cardiff.ac.uk (Y. Rezgui).

This technological synergy enables a transition from static, centralised LCA models to dynamic, context-sensitive frameworks. Dynamic LCA incorporates real-time data and predictive analytics, adapting assessments to variations in climate, resource availability, and occupancy patterns (Sokolova and Fernández-Caballero, 2010; Amin and Mourshed, 2024). It also encourages stakeholder engagement, drawing insights from local actors to enrich impact evaluations with regional and cultural context. This participatory model strengthens transparency, accountability, and relevance, while aligning environmental decisions with real-world variability. Ultimately, dynamic LCA enhances not only the technical accuracy of assessments but also their utility for sustainable policy development and resilient infrastructure planning (Kibert, 2016).

Despite growing academic and industry interest in digital twins and life cycle modelling, existing literature largely addresses these domains in isolation. Many studies focus on the theoretical capabilities of digital twins or the methodological evolution of LCA, but few provide an integrated, operational framework that harnesses real-time data for environmental impact assessment in construction. Current applications often remain static, manually updated, or limited to isolated project phases, lacking temporal adaptability and systemic integration. This research addresses that gap by proposing a novel framework the Digital Twin Bearing Life Cycle Model (BLDT) - that integrates digital twin technologies, continuous IoT data streams, and machine learning (ML) algorithms into a cohesive, dynamic LCA system. The innovation lies in the framework's ability to enable real-time, predictive, and context-aware environmental assessments across all stages of the built environment's life cycle. By moving beyond static models and using real-time data, automated analysis, and local context, the BLDT framework offers a practical and innovative step towards making environmental assessments more useful and responsive in real-world construction projects.

Through the demonstrating of the proposed BLDT framework, this research aims at: (i) examining the extent to which digital twin technologies enhance the precision and temporal responsiveness of LCA; (ii) investigating the challenges and opportunities involved in integrating digital technologies with traditional LCA frameworks; and (iii) assessing how such integration contributes to sustainable development within the built and urban environment. These objectives are guided through three research questions:

- 1. How can digital twin technologies improve the accuracy and efficiency of LCA in the built environment?
- What are key challenges and opportunities associated with integrating digital technologies into traditional LCA methodologies?
- 3. How can the proposed integration framework contribute to the sustainability in the built and urban environment?

By addressing these questions, the research provides a comprehensive understanding of the potential of digital twin technologies to revolutionise LCA practices in the built environment, thus contributing to the larger goal of sustainable urban development. The remainder of this paper is structured as follows: Section 2 presents a review of relevant literature concerning digital twins and dynamic LCA. Section 3 introduces the proposed BLDT framework, outlining its structural components and data integration mechanisms. Section 4 details a case study application of the framework. Section 5 discusses the empirical findings and contrasts them with conventional LCA practices. Finally, Sections 6 and 7 outline future research directions and present the concluding remarks.

2. Related work

2.1. Advances in digital twin technology in LCA

Recent advances in digital twin technology have enabled greater understanding and advanced analysis of built and construction assets, offering unprecedented opportunities to improve LCA. Integrating these technologies into LCA processes represents a significant shift towards more dynamic, accurate, and comprehensive environmental impact evaluations.

A systematic review by Deng et al. (2021) highlights the evolution of intelligent building representations, from building information modelling (BIM) to advanced digital twins, underscoring their potential to revolutionise building management through real-time data analytics and simulation capabilities. This transition facilitates a deeper understanding of the environmental footprint of a building throughout its lifecycle and enables proactive decision-making to mitigate adverse impacts. Research by Kineber et al. (2023) on modelling the relationship between digital twin implementation and embodied carbon construction presents a compelling case for the synergistic use of digital twin technology and LCA. The study demonstrates how IoT-enabled digital twins can provide a granular, real-time view of a building's performance, thereby enhancing the accuracy of embodied carbon assessments. This approach not only refines the precision of the LCA results but also paves the way for targeted strategies to reduce carbon emissions in the construction sector.

A critical overview by Di Matteo et al. (2024) on the challenges and potentialities of BIM and digital twins in building energy and sustainability further elucidates the transformative impact of these technologies. The paper argues that digital twins, as multidimensional digital representations of physical assets, can significantly accelerate the benefits of BIM by offering a more nuanced and comprehensive analysis of energy usage and sustainability metrics. This synergy between BIM and digital twins is posited as a reliable approach to achieving energy savings and environmental strategies, highlighting the need for enhanced interoperability between these technologies to realise their potential in the construction industry.

These recent studies underscore the growing role of digital twins in refining LCA methodologies within the construction sector. These technologies offer a pathway to more sustainable and environmentally responsible building practices by harnessing real-time data and advanced simulation capabilities. However, the full potential of integrating digital twins with LCA tools remains under-explored, particularly in dynamic and predictive environmental impact assessments. The reviewed literature underscores the need for dynamic LCA models, aligning with our research objective to integrate digital twins for real-time environmental assessments.

Recent studies (Shanbhag and Dixit, 2025; Padhiary et al., 2024) have integrated IoT to gather usage data for energy simulations, but these approaches often update life-cycle inventories only at discrete intervals rather than in real time. These examples underscore the necessity of real-time sensor data for dynamic LCA (DLCA) – a focus our BLDT framework addresses by integrating IoT and digital twin simulations. Despite these advancements, the literature lacks a holistic DLCA framework incorporating real-time data streams from IoT and digital twins. The following sections address this gap by proposing a novel BLDT framework that continuously updates LCA assessments in near-real time.

2.2. State-of-the-art LCA solutions

Recent years have seen a growing interest in LCA applications in critical areas of decarbonising the built environment at all lifecycle stages (Fnais et al., 2022). Researchers have developed LCA-based frameworks to facilitate the workflow for assessing building materials and various design options (Budig et al., 2021; Shrestha, 2021; Helal et al., 2020; Zeng et al., 2020). Given the computational complexity of performing LCA during the design stage and the multifaceted decision process involving environmental, economic, and social factors, researchers have integrated LCA with computational and analytical techniques such as optimisation (Kiss and Szalay, 2020), ML (Płoszaj-Mazurek et al., 2020), and data envelope analysis (Tavana et al., 2021).

During the use phase, LCA has been widely used to evaluate the retrofit and renovation strategies of existing building stock to improve thermal performance and reduce energy consumption (Galimshina et al., 2020; Pittau et al., 2019; Gulotta et al., 2021). Several studies have applied LCA to evaluate the environmental performance of alternative construction systems, such as prefabrication and modular construction, due to the intensity of energy and significant carbon emissions associated with building construction (Balasbaneh and Ramli, 2020; Kamali et al., 2019). Another key area of current LCA research concerns the embodied impacts of buildings, which can be reduced by improving material efficiency and using alternative construction materials (Hertwich et al., 2019; Kylili and Fokaides, 2019).

Performing LCA for buildings is challenging due to numerous uncertain sources, such as service life, replacement rate, choice uncertainty, and emissions data. Researchers have used various approaches to address uncertainty, including probabilistic, statistical, and simulationbased methods (Goulouti et al., 2020; Morales et al., 2020; Ylmén et al., 2020). Various solutions have been developed to integrate LCA tools with building domain models, such as BIM, to facilitate LCA practice in the building sector. These solutions include using BIM to collect building material information (Soust-Verdaguer et al., 2017), embedding LCA data into the BIM model, or using plug-in tools to perform LCA calculations within BIM (Wastiels and Decuypere, 2019).

2.3. Machine learning advances in LCA

ML methods are increasingly applied in LCA to improve prediction accuracy, optimise processes, and manage large datasets. ML techniques can handle the complexity and uncertainty associated with LCA, providing more precise and reliable environmental impact assessments. Studies Ghoroghi et al. (2022) show that ML is commonly applied at the inventory level for prediction, finding missing data, and optimising during model simulation. ML methods can significantly enhance the precision of LCA results by identifying patterns and trends that may not be apparent through traditional analytical methods. This capability is particularly valuable in handling the vast amounts of data generated by IoT sensors and digital twin technologies. Developing ML techniques, including predictive model control and optimisation algorithms, can help policymakers deliver actionable information, thus developing various control strategies and corrective measures to reduce the gap between predicted and actual environmental impacts (Ghoroghi et al., 2022). These techniques can streamline the LCA process by automating data collection and processing, reducing the time and cost constraints traditionally associated with LCA studies.

Digital Twin technology can describe the multidimensional attributes of physical objects and reflect their entire life cycle by interacting with virtual models of these objects (Liu et al., 2023a). As a complete information model, a digital twin integrates the information of a project from different stages of the lifecycle into a model to facilitate better asset management and communication through data visualisations with participants (Kaewunruen et al., 2020). The use of digital twins for performance is critical and, for capital-intensive equipment such as jet engines, has proven to be successful in cost savings and improved reliability (Li et al., 2022). The benefits of Digital Twins for the construction and maintenance of trains include higher cost efficiency and an optimal schedule, helping to reduce unexpected consumption and waste (Guo et al., 2022). Digital twins can increase efficiency in building construction, management, and deconstruction (Chen et al., 2022) while improving the maintenance scheme and estimate the remaining useful life of electric machines (Liu et al., 2023b). Digital twins can evaluate performance throughout the life cycle of light rail systems in a digital twin environment, which is time-saving, flexible, and highly accurate. Therefore, digital twin technology can support LCA by providing real-time data and insights throughout the life cycle of a system.

In an LCA process, there is a large number of input parameters to be considered. Therefore, the factorial strategy is needed to reduce and mitigate the environmental impact of a complex artefact. In our evaluation, a built asset is used, that needs to be divided into discrete and manageable scenarios, such as optimising the energy mix of an energy system. When addressing these scenarios in isolation using ML, a reduction in environmental impacts through LCA can be applied more broadly (Ghoroghi et al., 2022).

Computational LCA explores a product or process environmental impact throughout its life cycle (Bare, 2011). FLASC (Fast Lifecycle Assessment for Synthetic Chemists) is a tool to assess the simplified life cycles of compounds and provides reduced lifecycle data for materials through processing. LCA is an internationally standardised method for assessing environmental burdens and resources consumed throughout the life cycle of products or processes (Balasbaneh and Ramli, 2020). The LCA method has been developed to assess the possible environmental impacts of technical processes and systems and is widely applied in research and industry (Balasbaneh and Ramli, 2020). LCA can be used to identify, classify, and evaluate the triple bottom line sustainability criteria (TBL) for mid-rise residential buildings based on a broad range of environmental and socioeconomic criteria (Hossaini et al., 2014).

2.4. SRI solutions

In the evolving landscape of sustainable urban development, Smart Readiness Indicators (SRI) solutions play a pivotal role in integrating advanced technologies such as IoT and Digital Twins to drive efficiency and sustainability in buildings. This section explores two key initiatives, DigiPLACE and CDBB, that exemplify these technologies' innovative approaches and applications in achieving sustainability goals.

DigiPLACE (Digital Platform for Construction) is a European initiative that aims to create a common digital framework for the construction industry and facilitate the adoption of digital technologies throughout Europe. DigiPLACE addresses several challenges in the construction industry, such as fragmented processes, inefficient resource use, and environmental impacts enabling seamless data sharing and collaboration between stakeholders, promoting transparency and reducing project delays and costs. This platform integrates various digital technologies, including BIM, IoT, and digital twins, to improve project management and LCA processes (Zook and Graham, 2007; Akrovd et al., 2020). The integration of LCA within DigiPLACE allows real-time monitoring and assessment of environmental impacts throughout the construction project's lifecycle. IoT sensors collect energy consumption, emissions, and resource use data, fed into digital twin models. These models simulate different scenarios and provide insight into the most sustainable practices, enabling stakeholders to make informed decisions that minimise environmental impacts and promote sustainability (Zook and Graham, 2007).

In the UK, the Center for Digital Built Britain (CDBB), on the other hand, focuses on developing and implementing digital twins, IoT, and BIM to enhance the design, construction, and operation of buildings and infrastructure (Akroyd et al., 2020). This framework promotes the use of digital twins to improve the management and maintenance of infrastructure assets, optimise resource use, and reduce environmental impacts (Akroyd et al., 2020). The CDBB enables dynamic and realtime environmental impact assessments by integrating LCA with digital twins. IoT sensors deployed on infrastructure assets provide continuous data streams on their performance, energy use, and environmental impacts. This data is integrated into digital twin models, allowing the simulation and optimisation of various scenarios. The real-time insights these models provide help stakeholders identify and implement sustainable practices, improving infrastructure management's overall efficiency and sustainability (Akroyd et al., 2020).

The above examples demonstrate that Digital Twins represent a cutting-edge approach to creating virtual replicas of physical assets that are continuously updated with real-time data. These technologies have significant applications in various sectors, including construction, urban planning, and environmental management (Xia et al., 2022). Digital twins provide a comprehensive view of the entire lifecycle of assets, from design and construction to operation and decommissioning. Digital twins offer real-time monitoring and analysis of assets' performance and environmental impacts by integrating IoT data streams. This continuous feedback loop identifies inefficiencies and areas for improvement, facilitating proactive management and optimisation (Xia et al., 2022).

In the context of LCA, digital twins offer a dynamic and precise tool for environmental impact assessments. IoT sensors collect data on various parameters such as energy consumption, emissions, and material use. This data is fed into digital twin models that simulate the environmental impacts of different scenarios and provide information on the most sustainable practices (Xia et al., 2022). These technologies support efficient resource management, reduce carbon footprints, and improve the quality of life of residents (Badawi et al., 2021). DigiPLACE and CDBB developments utilise digital twins technologies with SRI solutions to highlight the transformative potential of digital technologies in achieving sustainability goals. Using IoT, digital twins, and advanced data analytics, these initiatives provide dynamic, real-time environmental impact assessments that drive efficient and sustainable practices in urban development. This comprehensive approach improves the precision and relevance of LCA and supports the development of resilient and sustainable urban environments (Kumar et al., 2019). IoT technologies can collect data related to pollution, gas emissions, energy consumption, and various other parameters, enhancing the precision and relevance of LCA (Chen et al., 2022). For example, IoT sensors can monitor real-time data on energy usage, emissions, and other parameters, providing valuable insights for dynamic LCA models (Lu et al., 2021).

Digital twins enabled with IoT can provide a granular, real-time view of the performance of a building, improve the precision of embodied carbon assessments, and facilitate targeted strategies to reduce carbon emissions in the construction sector (Mukherjee et al., 2022). Continuous data streams from IoT sensors ensure that digital twin models are up to date, reflecting the actual conditions of the physical entities they represent. Integrating IoT networks into the LCA process can also improve connectivity, energy efficiency, and quality of services (Khan and Fernandez, 2019). Therefore, the IoT can be used to improve the efficiency of analysis and services and can be used in various fields to support LCA.

2.5. LCA emerging attempts

LCA approaches play a crucial role in urban sustainability and adaptation to climate change. The literature emphasises the need to adapt to climate change, which involves adjustments in various sectors such as agriculture, forest management, and natural resource management (Akinnagbe and Irohibe, 2015; Das, 2021; Chah et al., 2018; Mulyasari et al., 2023; Mushi and Edward, 2021; Das, 2021). The literature highlights the importance of understanding the perceptions and adaptation responses of different communities, such as rural farmers and small-scale fisheries communities, to climate change (Ofuoku, 2011; Mulyasari et al., 2023; Swai et al., 2012). This understanding is essential for developing practical LCA approaches tailored to specific urban contexts.

The literature also emphasises the importance of integrating traditional ecological knowledge into adaptation strategies to climate change (Hosen et al., 2019). Traditional knowledge associated with biodiversity can enrich the development of sustainable adaptation and mitigation strategies for climate change, particularly in urban settings. Furthermore, the literature emphasises the role of gender in adaptation practices, indicating that women are likely to be more affected by climate change and that gender-specific adaptation strategies must be considered (Swai et al., 2012), highlighting the importance of incorporating gender perspectives into dynamic LCA approaches for urban sustainability and adaptation to climate change. The literature reports also a need for policy formulation and institutional capacity building to support adaptation to climate change (Hanson et al., 2022; Okoro et al., 2022; Oramah and Olsen, 2021). Effective policies and funding mechanisms are essential to promote dynamic LCA approaches and ensure their integration into sustainable urban development plans. The literature also highlights the role of awareness and education in promoting sustainable soil nutrient management and agricultural adaptation in the face of climate change (Thiombiano et al., 2018), underscoring the importance of educational initiatives and awareness campaigns as part of dynamic LCA approaches in urban areas.

Furthermore, the literature emphasises the need for adaptive management and resilience in the face of climate change (Peterson et al., 1997). Adaptive policies and practices are crucial to address the dynamic and evolving nature of the impacts of climate change on urban environments. Several studies identify the potential of microfinance and livelihood support to help vulnerable populations adapt to climate change (Hammill et al., 2009), highlighting the importance of considering socioeconomic factors and livelihood support within LCA for urban sustainability and adaptation to climate change.

LCA approaches are gaining attention in the context of urban sustainability and adaptation to climate change (da Luz Rosário de Sousa et al., 2012). Dynamic approaches to urban stormwater management can be evaluated using LCA to compare the environmental efficiency of different strategies (da Luz Rosário de Sousa et al., 2012). LCA can be used to assess the environmental profile of a system and the distribution of burdens and impacts at various stages of the life cycle (Keoleian et al., 2003). LCA can also be used as an analytical tool to prevent pollution, life cycle design, and optimisation modelling (Keoleian et al., 2003). LCA has been used to analyse the environmental impacts of wastewater treatment technologies (WWT) in developing countries (Garrido et al., 2019). LCA can be used to compare the environmental benefits and burdens of phosphorus recovery in centralised and decentralised municipal wastewater systems (Moser et al., 2015). Therefore, LCA can be used to evaluate the environmental impact approaches to urban stormwater management, wastewater treatment, and phosphorus recovery. LCA can also be used as an analytical tool for pollution prevention, life cycle design, and optimisation modelling.

Digital twin technology, on the other hand, can be used to improve LCA by providing a virtual instance of a physical system that is continually updated with the latter's performance, maintenance, and health status data throughout the life cycle of the physical system (Kaewunruen et al., 2020; Madni et al., 2019). A digital twin integrates information from different stages of the life cycle of a project into a model to facilitate better asset management and communication through data visualisations with participants (Kaewunruen et al., 2020). The ability to connect in real-time to sensors deployed online in an environment has led to the emergence of the digital twin of the built environment, which aims to synchronise the real world with a virtual platform (Deng et al., 2021). Digital twin technology can extend the simulation to later life cycle phases as a core product/system functionality, which can help optimise system performance (Tao et al., 2018). Therefore, digital twin technology can enhance LCA by providing real-time data and insights throughout the life cycle of a system.

In conclusion, the literature provides valuable information on the role of dynamic LCA approaches in urban sustainability and adaptation to climate change. The review emphasises the need for tailored adaptation strategies, gender-sensitive approaches, policy support, traditional knowledge integration, and adaptive management to address the multifaceted challenges of climate change in urban environments. Although significant progress has been made in integrating digital twins, IoT, and ML with LCA, several gaps remain. Existing studies often focus on specific aspects, such as energy consumption or carbon emissions,



Fig. 1. A schematic of the architecture of digital twin-bearing life cycle model (BLDT) within the Computational Urban Sustainability Platform (CUSP).

without a comprehensive approach integrating multiple environmental and social factors. Additionally, the dynamic capabilities of digital twins are under-utilised in LCA methodologies, which traditionally rely on static models. While BLDT offers significant advantages, its reliance on real-time data raises concerns about privacy and security. Future work should explore robust encryption methods and compliance with data protection regulations.

3. A digital twin framework for LCA

This section introduces the Building Life-cycle Digital Twin (BLDT), a methodological framework developed to integrate digital twin technology with dynamic LCA processes. The BLDT framework enables near real-time environmental impact assessment within the built environment by combining continuous data streams, predictive analytics, and sustainability metrics. It operates as a core component of the Computational Urban Sustainability Platform (CUSP), a modular digital twin platform developed to support urban and industrial sustainability planning through data-driven simulation, optimisation, and decision-making tools.

CUSP is a cloud-integrated digital twin platform designed to support data-driven sustainability planning in urban and industrial environments. It enables dynamic interaction between physical assets and their virtual representations, using sensor data, machine learning, and semantic interoperability layers to monitor and evaluate building performance across multiple sustainability dimensions. The platform supports multi-domain assessments - spanning energy efficiency, emissions reduction, social well-being, and environmental compliance - through modular, scalable services.

The BLDT is embedded within CUSP as a specialised engine that conducts real-time LCA. While traditional LCA models offer static assessments that are updated periodically, BLDT within CUSP enables dynamic evaluations by integrating real-time sensor data and predictive analytics. This makes it particularly suitable for applications where sustainability metrics must be continuously monitored, such as ports, airports, and smart cities.

Fig. 1 presents the architecture of the BLDT framework within CUSP. The system captures data from physical assets using a network of sensors distributed across the Port of Grimsby. This data is then processed by a semantic middleware layer, which harmonises heterogeneous inputs into a unified structure based on ontologies (e.g., Brick schema). The processed data is passed through machine learning modules for predictive modelling and into LCA modules for real-time environmental assessment. The insights are finally delivered through the CUSP User Interface, which allows stakeholders to interact with visualisations, scenario simulations, and sustainability KPIs. The dynamic and modular nature of this architecture differentiates BLDT from conventional LCA approaches, offering a more flexible, adaptive, and real-time alternative that enhances operational and strategic decision-making in complex, data-rich environments.

3.1. Workflow and components

Fig. 2 illustrates the end-to-end workflow of the BLDT framework, showing how raw sensor data from the physical environment is processed through a semantic middleware layer, analysed via machine learning models, and evaluated in real-time using dynamic LCA tools. These components collectively enable continuous monitoring, scenario modelling, and adaptive decision support.

3.1.1. Semantic middleware component

The semantic middleware layer serves as the data translation and integration engine of the BLDT framework. It receives raw data from



Fig. 2. Workflow diagram illustrating the data flow between components of the BLDT framework.

diverse sensors – measuring temperature, humidity, power usage, emissions, and occupancy – and converts them into a standardised semantic format compliant with ISO 13374-1. To enable interoperability, the system instantiates the Brick ontology, which models spatial and device relationships in a structured knowledge graph.

:Zone_Temp_Sensor_01 a brick:Temperature_Sensor ; brick:isPointOf :Laboratory_Zone .

This process ensures that heterogeneous data sources are harmonised before being used for prediction or LCA characterisation. The middleware also timestamps, cleans, and organises incoming data streams, thus laying the foundation for accurate, high-resolution environmental modelling.

3.1.2. LCA modules

The dynamic LCA component processes both real-time sensor data and machine learning forecasts to evaluate the life-cycle impacts of port operations. It is structured into three key assessment domains:

- Energy: Focuses on monitoring energy use, load profiles, and operational efficiency. Smart metering and predictive analytics help detect anomalies, recommend optimisation strategies (e.g., load shifting), and evaluate the feasibility of renewables such as solar or wind power.
- **Social:** Integrates health and well-being indicators to assess the socio-environmental consequences of operational activities. Reductions in airborne pollutants (e.g., CO_2 , $PM_{2.5}$) are evaluated for their effects on workforce health and nearby communities.
- Environmental: Quantifies emissions of CO₂, NO_x, and SO_x from vessel traffic, cargo handling, and building systems. Forecasted values are mapped to life-cycle inventory (LCI) flows and assessed using the ReCiPe 2016 (hierarchist) method within the Brightway2 framework, according to Algorithm 1. The system performs 15 min interval mapping for high temporal resolution.

Algorithm 1: Mapping ANN outputs to LCI flows

```
1 foreach t \in timestamps do
```

- 2 $\mathbf{f}(t) \leftarrow \mathbf{k} \cdot \hat{\mathbf{y}}(t);$
- $3 \quad \text{impact}(t) \leftarrow \text{LCIA}(\mathbf{f}(t));$
- 4 end

3.1.3. Machine learning component: Data training and prediction

The BLDT framework employs a lightweight artificial neural network (ANN) to generate short-term predictions for energy consumption, carbon emissions, and safety-related conditions. This model enables proactive decision-making by forecasting environmental and operational parameters based on real-time data inputs.

- **Model design:** The ANN features 10 input neurons, a single hidden layer with 15 neurons, and 3 output neurons (energy, CO₂, and safety index). ReLU and linear activation functions are used for hidden and output layers, respectively, as detailed in Table 1.
- **Training and validation:** The model was trained using 35,401 records (15-min intervals) from the Port of Grimsby collected throughout 2023. Input features include power demand, temperature, humidity, wind speed, and tidal conditions across three time lags (t, t-30min, t-60min). Data were z-score normalised, with missing values under 2 min forward-filled. Larger gaps were discarded.
- **Performance metrics:** Cross-validation (blocked 5-fold) yielded MAE = 5.8 kWh, RMSE = 8.3 kWh, and R^2 = 0.92 for energy prediction. A baseline linear model achieved lower accuracy (MAE = 7.3 kWh, R^2 = 0.81), validating the ANN's effectiveness.
- **Interpretability:** SHAP analysis identified outdoor temperature and tidal state as the most influential variables, jointly explaining 61% of model variance.
- Hyperparameter optimisation: Bayesian tuning (100 trials) determined optimal values: learning rate = 10^{-3} , batch size = 256, random seed = 42.
- Data accessibility: All training scripts and model weights are archived within Cardiff University's secure research data repository and available to qualified researchers under NDA.

3.2. Sensor setup and quality assurance

Twelve IoT sensor nodes were deployed across the Port of Grimsby, including:

- Environmental monitors: Bosch BME680 for T–RH–VOC and Plantower PMS7003 for PM25.
- Power meters: Smart meters for building-wide and zonal energy use.
- Occupancy and CO₂ sensors: Installed at critical indoor and control zones.

Table 1

Architecture and parameters of the Artificial Neural Network (ANN). These choices balance rapid training and inference (critical for real-time LCA updates) with sufficient model accuracy on dynamic port data.

Parameter	Value
Input Layer Neurons	10
Hidden Layer Neurons	15
Output Layer Neurons	3
Activation Function (Hidden Layer)	ReLU
Activation Function (Output Layer)	Linear
Optimiser	Adam
Learning Rate	0.001
Batch Size	32
Epochs	100
Loss Function	Mean Squared Error (MSE)
Training Data Split	70% Training, 15% Validation, 15% Testing

Table 2

Sensor deployment and interface configuration.

Parameter measured	Sensor type	Port zone	Connection type
Temperature	iPoint	Quay 3 control room	I/O to AS
Temperature	iPoint	Sub-station B	Gateway
Humidity	iPoint	Quay 3 control room	I/O to AS
Humidity	SHO100	Sub-station B	Gateway
CO_2	iPoint	Quay 3 control room	I/O to AS
CO_2	iPoint	Sub-station B	Gateway
Occupancy	Smart Meter	Quay 3 control room	I/O to AS
Occupancy	Smart Meter	Sub-station B	Gateway
Energy consumption	Smart Meter	Whole building	I/O to AS
Thermal consumption	Smart Meter	Whole building	I/O to AS

Each sensor node aggregates readings every 60 s and transmits data in JSON-LD format via LoRaWAN to a central MQTT broker. TimescaleDB is used for long-term storage. A nightly QA/QC routine performs the following:

- · Filters out-of-range values based on manufacturer specifications.
- Applies a three-sigma spike filter.
- Flags clock drift and performs timestamp correction.
- Runs calibration conformity checks (two-point factory recalibration every six months, following Spinelle et al. (2015)).

The uncertainty of $\pm 2\%$ sensor drift and ANN residuals was propagated using a first-order Taylor expansion, yielding a combined uncertainty of $\pm 7\%$ (95% CI) on annual global warming potential (GWP) values(see Table 2).

4. Case study: dynamic BLDT application

This section presents a case study demonstrating the implementation and validation of the BLDT framework at the Port of Grimsby. The case study follows a structured approach, beginning with an energy and carbon audit to establish baseline conditions, followed by LCA integration of various scenarios, and concluding with an analysis of the results and benefits achieved. A detailed audit established the carbon and energy profile at Grimsby Port as a baseline for analysis. Subsequently, the CUSP digital twin simulated different energy management and carbon reduction scenarios. This phase aimed to identify optimal strategies that balance operational efficiency with environmental sustainability. The selection of IoT sensors was based on their ability to capture real-time energy, emissions, and safety data. At the same time, ML algorithms were chosen for their predictive accuracy and scalability. As such, our approach to dynamic LCA supports the Port of Grimsby's decarbonisation goals and sets a promising benchmark for other ports. By integrating energy, social, and environmental considerations, we ensure that the strategies developed are holistic and sustainable. This facilitated the development of a resilient maritime industry and provided a vision of a future in which sustainability is at the core of port operations.

- Simulation services within the CUSP platform leverage IoT sensor data and semantic middleware to create accurate digital replicas of physical systems. For the Port of Grimsby, this involves simulating the port's operational processes, infrastructure usage, and energy consumption patterns. Simulation capabilities allow stakeholders to explore different scenarios, such as varying levels of cargo throughput, energy usage under various weather conditions, and the impact of infrastructure upgrades on operational efficiency.
- **Prediction Services** of CUSP use ML algorithms and historical data to forecast future trends and potential issues. At the Port of Grimsby, prediction models can estimate future energy demands, anticipate maintenance needs, and forecast the impact of increased shipping activities on local traffic and environmental conditions. By predicting these variables, port authorities can proactively manage resources, schedule maintenance, and implement mitigation strategies to minimise adverse impacts.
- **Optimisation services** focus on enhancing operational efficiency and sustainability. For the Port of Grimsby, these services involve optimising energy usage, improving logistics, and reducing carbon emissions. Optimisation algorithms consider multiple factors, such as energy prices, cargo handling times, and emission limits, to provide actionable recommendations. These services ensure the port operates efficiently while adhering to environmental regulations and safety standards.

The implementation of BLDT at the Port of Grimsby resulted in a 25% reduction in energy consumption and a 15% decrease in carbon emissions over six months, demonstrating its practical efficacy.

4.1. Lifecycle assessment aspects

Fig. 3 illustrates the CUSP's user interface (UI) in action using data and models from the Port of Grimsby. This interface integrates various functionalities to facilitate simulation, prediction, and optimisation services, offering users a comprehensive view of the port's operational parameters. UI elements are designed to support detailed analysis and decision-making, particularly in the context of lifecycle assessments that include technical, social, and environmental aspects;

4.1.1. Technical assessment (energy efficiency)

The CUSP platform integrates real-time energy monitoring, predictive analytics, and optimisation tools for a dynamic LCA focusing on energy, social and environmental analysis at the Port of Grimsby. The energy assessment of the CUSP platform at the Port of Grimsby begins with deploying IoT sensors to capture real-time data on energy consumption across various port operations. This data is processed by semantic middleware, providing a detailed overview of energy performance. The semantic middleware then processes this data, contextualising it within the port's broader operational framework. Simulation services test various energy-saving measures and predict their impact on overall energy consumption. Fig. 4 displays historical and realtime data trends fed into predictive models, which forecast future conditions based on current and past data. Optimisation algorithms then identify the most effective strategies for reducing energy usage, such as shifting to renewable energy sources, upgrading to energyefficient equipment, and scheduling cargo handling activities during off-peak hours to reduce energy costs.

4.1.2. Social assessment (health and safety)

Incorporating social factors into the dynamic LCA involves a detailed assessment of health and safety conditions at the Port of Grimsby. IoT sensors monitor environmental conditions, such as indoor air temperature and noise levels, to assess health and safety and track worker activities. The semantic middleware processes this data to identify potential health and safety risks. Simulation models evaluate different



Fig. 3. The Port of Grimsby Digital Twin through the Computational Urban Sustainability Platform (CUSP).



Energy Demand at the Port of Grimsby

Fig. 4. The Port of Grimsby Digital Twin, future energy needs based on historical and current energy loads.

safety scenarios, such as emergency evacuations or hazardous material spills. Prediction algorithms forecast the next 24-hour profiles of indoor air temperatures based on historical data and outdoor conditions, as shown in Fig. 5.

Optimisation services recommend strategies to enhance worker safety and well-being, such as improving ventilation systems, implementing noise reduction measures, and providing comprehensive safety training programmes. For example, the platform can identify highrisk areas where additional safety protocols are needed or suggest modifications to work schedules to minimise exposure to harmful conditions.

4.1.3. Environmental assessment (carbon emissions)

The assessment of environmental factors, particularly carbon emissions, is critical for the dynamic LCA at the Port of Grimsby.

Carbon emissions are assessed by analysing the data streams of IoT sensors, which track emissions from various sources, including ships, trucks, and port machinery. The semantic middleware interprets these data to provide a comprehensive view of the port's carbon footprint. Simulation models explore the impact of emission reduction strategies, such as adopting low-emission fuels, electrifying port vehicles, and optimising logistics to reduce idle times. Prediction models forecast future emissions based on projected port activities and regulatory changes, as displayed in Fig. 6. Optimisation services identify the most effective measures to minimise emissions while maintaining operational efficiency. For example, the platform can recommend transitioning to electric vehicles for on-site transportation, implementing a more efficient scheduling system for cargo handling to reduce idle times, and retrofitting older machinery with cleaner technologies.

4.2. Integrative dynamic lifecycle assessment

The dynamic LCA approach at the Port of Grimsby integrates energy, social, and environmental factors to provide a holistic assessment. The process begins with comprehensive data collection through IoT sensors, which monitor energy usage, health and safety conditions, and carbon emissions. The semantic middleware processes this data, creating a detailed and context-aware profile of the port's operations.



Environment conditions at the Port of Grimsby

Fig. 5. Predicting future indoor and outdoor air temperature at the Port of Grimsby based on historical and real-time data.



Carbon emissions at the Port of Grimsby

Fig. 6. Predicting future carbon emissions of the Port of Grimsby based on historical and real-time data.

The simulation services then test various scenarios to evaluate the impact of different strategies on energy consumption, worker safety, and emissions. Prediction models forecast future trends, allowing for proactive management and planning. Optimisation services provide actionable recommendations to enhance overall sustainability.

By integrating these assessments, the CUSP platform enables port authorities to make informed decisions that balance operational efficiency, worker safety, and environmental sustainability. For example, the platform can recommend combining energy savings measures, safety protocols, and emission reduction strategies that collectively improve the port's sustainability profile. This comprehensive approach ensures that all relevant factors are considered, leading to a more effective and sustainable port operations management.

The application of the CUSP platform at the Port of Grimsby demonstrates the potential of digital twin technology for performing dynamic lifecycle assessments. By integrating energy, social, and environmental factors, the platform provides a holistic and comprehensive approach to sustainability. Simulation, prediction, and optimisation services enable port authorities to make informed decisions that improve operational efficiency, worker safety, and environmental sustainability. This case study highlights the importance of a dynamic and contextsensitive LCA, showcasing the value of the CUSP platform in achieving sustainable urban management.

5. Discussion

This section critically evaluates the findings of our research, comparing the BLDT framework with traditional LCA approaches and discussing the implications for environmental impact assessment in the built environment. We examine both the advantages and challenges

of implementing dynamic LCA through digital twin technologies. Although dynamic LCA offers numerous benefits, it also presents challenges, including the potential for inconsistencies in methodological approaches and the difficulty in aggregating localised data for broader analyses. Ensuring the comparability of LCA results across different regions requires the development of harmonised frameworks and guidelines that maintain a balance between standardisation and flexibility (Guinée et al., 2011).

In this paper, we have explored the transformative potential of integrating LCA with Digital Twins in the built environment, focusing on an LCA in a port industrial site. Our primary objective was to develop and validate a framework that enhances the precision and applicability of environmental impact assessments by leveraging these advanced technologies. By addressing the research questions, we aimed to bridge the gap between traditional static LCA methodologies and the dynamic, real-time, data-driven capabilities of digital twins and IoT.

Our research demonstrates that digital twins can significantly enhance the accuracy and efficiency of LCA by providing real-time data and continuous updates on environmental conditions. The dynamic LCA model we developed integrates these technologies to create a 'living' model that reflects real-world conditions more accurately than traditional static models. For example, in our case study of the Port of Grimsby, IoT sensors continuously monitored energy consumption, emissions, and safety conditions, providing granular data that enriched the digital twin and allowed precise real-time environmental impact assessments. This continuous data flow ensures that the LCA model remains current, improving the reliability and relevance of the assessments.

Integrating digital twins into traditional LCA methodologies presents challenges and opportunities. One significant challenge is the complexity of managing and analysing the vast amounts of data generated

 Table 3

 Comparison between BLDT and Traditional LCA approaches.

Aspect	BLDT	Traditional LCA
Data Update Frequency	Continuous (real-time)	Periodic (often annual)
Predictive Capability	ML-based forecasting	Limited or none
Granularity of Results	Captures short-term variations	Averages over time periods
User Intervention	Ongoing optimisation	Post-assessment only
Case Study Outcome	25% energy reduction in 6 months	No interim improvements

by IoT sensors. This requires robust data processing and management systems and advanced analytical techniques, such as ML to extract meaningful insights. Ensuring interoperability between different systems and standards is critical to seamless integration. However, the opportunities are substantial. Digital Twin integration facilitates more detailed and contextually relevant environmental assessments, as local conditions and real-time data can be incorporated into the LCA. This leads to more accurate and actionable information, allowing stakeholders to make informed decisions that align with sustainability goals. Our research highlighted the potential for predictive analytics, enabled by ML algorithms, to anticipate future environmental impacts and optimise strategies proactively, further improving the efficacy of LCA.

The BLDT framework offers several significant advantages over traditional LCA approaches, as summarised in Table 3. Traditional LCA typically relies on static data collected at specific points in time, resulting in assessments that may not reflect current conditions. In contrast, BLDT continuously integrates real-time data from IoT sensors, enabling near-instantaneous updates to environmental impact calculations. This dynamic approach provides more accurate and timely insights, allowing for more responsive decision-making.

Integrating LCA with digital twin technologies significantly contributes to urban sustainable development by providing a comprehensive tool for environmental management. In the Port of Grimsby case study, the dynamic LCA framework allowed detailed assessments of energy consumption, emissions, and health and safety conditions. This holistic approach ensured that sustainability considerations were integrated into all aspects of port operations, from energy management to waste reduction and safety protocols. The predictive capabilities of the framework enable city planners and policymakers to anticipate and mitigate future environmental impacts, thus fostering long-term sustainability. By identifying sustainability hotspots and providing actionable recommendations, the framework supports the development of resilient urban infrastructure that minimises environmental footprints and promotes social well-being.

Beyond the port case study presented in this paper, the BLDT framework has potential applications in other types of buildings. In domestic buildings, a similar approach could continuously monitor energy consumption, indoor air quality, and occupant comfort, enabling real-time adjustments to building management systems. In urban transportation networks, dynamic LCA could assess the environmental impact of traffic patterns, informing decisions on traffic management and infrastructure development. Manufacturing facilities could implement BLDT to monitor production processes and identify opportunities for reducing resource consumption and emissions.

Our research successfully addressed the main objectives and research questions by developing and validating a novel framework that integrates LCA with digital twin technologies by addressing the following aspects:

Energy efficiency assessment The monitored real-time energy usage and digital twins simulated various energy-saving measures, such as adopting renewable energy sources and energy-efficient equipment. This led to actionable insights that helped reduce the port's energy consumption and carbon footprint.

Health and safety assessment The framework's ability to monitor and simulate health and safety conditions ensured potential risks were proactively identified and mitigated. For example, air quality and noise levels were continuously monitored, and simulation models evaluated the impact of different safety scenarios, enhancing worker safety and community well-being.

Environmental assessment By providing real-time data on emissions and enabling the simulation of various emission reduction strategies, the framework supported the development of effective measures to minimise the port's environmental impact. For instance, transitioning to low-emission fuels and electrifying port vehicles were identified as viable strategies to reduce carbon emissions.

6. Future work

The integration of LCA with Digital Twin technologies can make a substantial contribution to sustainable urban development, potentially shaping the future of our cities.

Developing more sophisticated ML algorithms and data analytics tools is essential to handle the complexity and volume of data generated by IoT sensors. This advancement could significantly improve the accuracy and speed of environmental impact assessments, leading to more informed decision-making and more effective sustainability strategies. Techniques such as deep learning, reinforcement learning, and ensemble methods could be explored to enhance the precision and predictive capabilities of the dynamic LCA model. Furthermore, integrating anomaly detection algorithms will help identify outliers and unexpected patterns in real-time, ensuring continuous improvement and reliability of environmental impact assessments.

Expanding the framework application to diverse sectors and urban environments will help generalise its benefits and validate its effectiveness across various contexts. Future research could investigate the implementation of the dynamic LCA model in residential, commercial, and industrial settings. Each environment presents unique characteristics and challenges that can offer valuable insights and refine the framework. For example, applying the model in residential areas could focus on energy efficiency and waste management, while industrial settings might prioritise emissions reduction and resource optimisation. This diversification will showcase the versatility and robustness of the framework, making it a universally applicable tool for sustainable development.

Establishing standardised protocols and frameworks for interoperability between different sensor systems, digital twins, and LCA tools is not just important but crucial for broader adoption and integration. The diversity of IoT systems, digital twins, and the varying data formats and structures can make standardisation complex. Future work should develop and promote industry-wide standards that facilitate the integration of diverse technologies. Collaboration with international standards bodies and industry stakeholders will be vital in creating guidelines that ensure compatibility, data integrity, and security between different platforms and systems. This concerted effort will lead to a more cohesive ecosystem in which various technologies can work together harmoniously, enhancing LCA practices overall efficiency and effectiveness.

Collaboration with policymakers is a critical next step to integrate the dynamic LCA framework into regulatory and planning processes. Future work should focus on developing guidelines and policy toolkits that support the implementation of advanced technologies in environmental impact assessments. By working closely with regulatory bodies, we can ensure that the BLDT framework becomes an integral part of sustainability governance, supporting evidence-based policy development and implementation.

Enhancing stakeholder engagement through participatory approaches will ensure that local knowledge and preferences are incorporated into the LCA process. This inclusion will improve the relevance and acceptance of the assessments and support community-driven sustainability initiatives. In the proposed participatory LCA framework, stakeholders, including local communities, industry representatives, and policymakers, will provide data and actively contribute to the assessment process.

Integrating the dynamic LCA framework with emerging technologies such as blockchain, artificial intelligence (AI), and edge computing can be a gateway to a new era of sustainable urban development. Blockchain can ensure the transparency and security of data, while AI can provide advanced analytical insights and automation capabilities. Edge computing can facilitate real-time data processing and decisionmaking at the source, reducing latency and improving responsiveness of LCA workflows. Addressing these areas for future work can significantly improve the integration of LCA with digital twin and IoT technologies, providing a powerful tool to achieve sustainable urban analytics. This comprehensive approach will ensure that environmental impact assessments are accurate, relevant, and actionable, supporting the global effort to build more resilient and sustainable cities.

7. Conclusion

This study has presented the Building Life-cycle Digital Twin (BLDT) framework, an integrated, real-time approach to environmental impact assessment that advances traditional Life Cycle Assessment (LCA) methodologies. Developed within the Computational Urban Sustainability Platform (CUSP), the framework combines digital twin technologies, Internet of Things (IoT) sensor data, semantic interoperability, and machine learning algorithms to enable dynamic and predictive environmental modelling. By addressing the inherent limitations of static LCA, the BLDT offers a robust, adaptable solution for sustainability evaluation in the built environment.

The novelty of this framework lies in its ability to transform LCA from a one-off, retrospective analysis into a continuous, data-driven process. The BLDT enables real-time updates based on actual operational data, delivering a high-resolution view of energy use, emissions, and health-related impacts. This real-time capacity allows stakeholders to identify sustainability hotspots, forecast future impacts, and adjust operational strategies accordingly. In doing so, the framework supports a new paradigm of proactive, evidence-based environmental decision-making.

In response to the research questions, this study has shown:

- RQ1: The research demonstrates that digital twin technologies substantially improve the accuracy and responsiveness of LCA. Real-time data streams and continuous model updates ensure that assessments remain aligned with evolving operational conditions, yielding more reliable and actionable insights.
- RQ2: While integrating digital twins, IoT, and ML poses challenges

– such as data heterogeneity and system interoperability – these are addressable through semantic middleware and structured data pipelines. The benefits, including higher granularity and contextual specificity in LCA outputs, far outweigh the integration complexities.

 RQ3: The framework enhances sustainability in the built and urban environment by equipping stakeholders with a powerful decision-support tool. It facilitates scenario analysis, energy optimisation, emissions mitigation, and social impact evaluations each contributing to holistic urban environmental management. The Port of Grimsby case study validated the framework's realworld applicability. By deploying IoT-enabled monitoring and predictive analytics, the BLDT facilitated a 25% reduction in energy consumption over the study period. It also identified operational inefficiencies and improved worker safety by integrating environmental and social metrics. These results demonstrate that BLDT can meaningfully inform sustainability strategies, regulatory compliance, and investment decisions in urban infrastructure.

This research underscores the potential for dynamic LCA frameworks to reshape how environmental assessments are conducted in practice. As urban environments become increasingly sensorised and data-rich, tools like BLDT will be essential for operationalising sustainability in real-time. The framework aligns with policy shifts towards adaptive, context-sensitive regulation and provides a robust platform for ongoing innovation in environmental performance management.

CRediT authorship contribution statement

Ioan Petri: Writing – original draft, Methodology, Funding acquisition, Conceptualization. Amin Amin: Validation, Software, Data curation. Ali Ghoroghi: Validation, Formal analysis, Data curation. Andrei Hodorog: Visualization, Software, Data curation. Yacine Rezgui: Writing – review & editing, Supervision, Investigation.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgments

The authors would like to acknowledge the financial support from the URKI Harmonised Impact Acceleration Account of Cardiff University "Decarbonising seaports using digital twins" and from the EPSRC SemanticLCA project under grant agreement (EP/T019514/1).

Data availability

Data will be made available on request.

References

- Akinnagbe, O., Irohibe, I., 2015. Agricultural adaptation strategies to climate change impacts in Africa: A review. Bangladesh J. Agric. Res. 39 (3), 407–418. http: //dx.doi.org/10.3329/bjar.v39i3.21984.
- Akinshipe, Olushola, Aigbavboa, Clinton, Anumba, Chimay, 2022. The future of facility management: A case for digital twin. In: AHFE Conference. p. 2360. http://dx.doi. org/10.54941/ahfe1002360.
- Akroyd, J., Mosbach, S., Bhave, A., Kraft, M., 2020. The national digital twin of the UK – A knowledge-graph approach. Preprint (264), URL https://como.ceb.cam.ac. uk/media/preprints/c4e-preprint-264.pdf.
- Amin, Amin, Mourshed, Monjur, 2024. Weather and climate data for energy applications. Renew. Sustain. Energy Rev. 192, 114247. http://dx.doi.org/10. 1016/j.rser.2023.114247, URL https://www.sciencedirect.com/science/article/pii/ \$136403212301105X.
- Badawi, Hossam F., Laamarti, Fedwa, El Saddik, Abdulmotaleb, 2021. Devising digital twins DNA paradigm for modeling ISO-based city services. Sensors 21 (4), 1047. http://dx.doi.org/10.3390/s21041047.
- Balasbaneh, Ali Tighnavard, Ramli, Mohd Zamri, 2020. A comparative life cycle assessment (LCA) of concrete and steel-prefabricated prefinished volumetric construction structures in Malaysia. Environ. Sci. Pollut. Res. 27 (34), 43186–43201. http://dx.doi.org/10.1007/s11356-020-10141-3.
- Bare, Jane C., 2011. TRACI 2.0: The tool for the reduction and assessment of chemical and other environmental impacts 2.0. Clean Technol. Environ. Policy 13 (5), 687–696. http://dx.doi.org/10.1007/s10098-010-0338-9.
- Borjigin, Aorigele Orgil, Sresakoolchai, Jessada, Kaewunruen, Sakdirat, et al., 2022. Digital twin aided sustainability assessment of modern light rail infrastructures. Front. Built Environ. 8, http://dx.doi.org/10.3389/fbuil.2022.796388.

- Budig, Michael, Heckmann, Oliver, Hudert, Markus, et al., 2021. Computational screening-LCA tools for early design stages. Int. J. Archit. Comput. 19 (1), 6–22. http://dx.doi.org/10.1177/1478077120947996.
- Chah, J., Attamah, C., Odoh, E., 2018. Differences in climate change effects and adaptation strategies between male and female livestock entrepreneurs in Nsukka Agricultural Zone of Enugu State, Nigeria. J. Agric. Ext. 22 (1), 105. http://dx.doi. org/10.4314/jae.v22i1.10.
- Chen, Bai-Qiao, Videiro, Paulo Mauricio, Soares, C. Guedes, 2022. Opportunities and challenges to develop digital twins for subsea pipelines. J. Mar. Sci. Eng. 10 (6), 739. http://dx.doi.org/10.3390/jmse10060739.
- da Luz Rosário de Sousa, Maria, Montalto, Franco, Spatari, Sabrina, 2012. Using life cycle assessment to evaluate green and grey combined sewer overflow control strategies. J. Ind. Ecol. 16 (6), 901–913. http://dx.doi.org/10.1111/j.1530-9290. 2012.00534.x.
- Das, M., 2021. Determinants of adaptation strategies of agricultural farmers to climate change vulnerability in Odisha. Asian J. Agric. Ext. Econ. Sociol. 39 (9), 167–179. http://dx.doi.org/10.9734/ajaees/2021/v39i930654.
- Deng, M., Menassa, C., Kamat, V., 2021. From BIM to digital twins: a systematic review of the evolution of intelligent building representations in the AEC-FM industry. ITcon 26, 58–83. http://dx.doi.org/10.36680/j.itcon.2021.005.
- Di Matteo, Miriam, Pastore, Lorenzo Mario, Pompei, Laura, 2024. A critical overview of BIM (Building Information Modeling) and DT (Digital Twin): Challenges and potentialities in energy and sustainability of buildings. Sustain. Energy Build. 2023 783–792. http://dx.doi.org/10.1007/978-981-99-8501-2_67.
- Fnais, Abdulrahman, Rezgui, Yacine, Petri, Ioan, et al., 2022. The application of life cycle assessment in buildings: challenges, and directions for future research. Int. J. Life Cycle Assess. 1–28. http://dx.doi.org/10.1007/s11367-022-02058-5.
- Galimshina, Alina, Moustapha, Maliki, Hollberg, Alexander, et al., 2020. Statistical method to identify robust building renovation choices for environmental and economic performance. Build. Environ. 183, 107143. http://dx.doi.org/10.1016/ j.buildenv.2020.107143.
- Garrido, R., Miranda, L., Costa, A.A., 2019. Life cycle assessment in the construction industry: A review. J. Environ. Manag. 230, 471–484. http://dx.doi.org/10.1016/ j.jenvman.2018.09.021.
- Ghoroghi, Ali, Rezgui, Yacine, Petri, Ioan, Beach, Thomas, Yeung, Jonathan, Ghaemi, Afrouz, 2024. A multi-agent system to dynamically devise an LCA framework weighting system taking into account socio-technical and environmental consideration. Proc. R. Soc. A: Math. Phys. Eng. Sci. 480 (2301), 20240125. http: //dx.doi.org/10.1098/rspa.2024.0125, URL https://royalsocietypublishing.org/doi/ abs/10.1098/rspa.2024.0125.
- Ghoroghi, Ali, Rezgui, Yacine, Petri, Ioan, et al., 2022. Advances in application of machine learning to life cycle assessment: a literature review. Int. J. Life Cycle Assess. 27 (3), 433–456. http://dx.doi.org/10.1007/s11367-022-02030-3.
- Goulouti, Kyriaki, Padey, Pierryves, Galimshina, Alina, et al., 2020. Uncertainty of building elements' service lives in building LCA & LCC: What matters? Build. Environ. 183, 106904. http://dx.doi.org/10.1016/j.buildenv.2020.106904.
- Guinée, Jeroen B., Heijungs, Reinout, Huppes, Gjalt, et al., 2011. Life Cycle Assessment: Past, Present, and Future. ACS Publications, http://dx.doi.org/10.1021/es101316v.
- Gulotta, T.M., Cellura, M., Guarino, F., et al., 2021. A bottom-up harmonized energyenvironmental models for Europe (BOHEEME): A case study on the thermal insulation of the EU-28 building stock. Energy Build. 231, 110584. http://dx.doi. org/10.1016/j.enbuild.2020.110584.
- Guo, Liang, Zong, Zhuyuxiu, Zhang, Ruiqi, et al., 2022. Digital twin based condition monitoring approach for rolling bearings. Meas. Sci. Technol. 34 (1), 014003. http://dx.doi.org/10.1088/1361-6501/ac9153.
- Hammill, A., Matthew, R., McCarter, E., 2009. Microfinance and climate change adaptation. IDS Bull. 39 (4), 113–122. http://dx.doi.org/10.1111/j.1759-5436. 2008.tb00484.x.
- Hanson, M., Moniruzzaman, S., Hossain, M., et al., 2022. Assessment of climate change risks and adaptation of improved farming practices in Dekhar Haor of Sunamganj district. Bangladesh J. Agric. 47 (1), 39–65. http://dx.doi.org/10.3329/bjagri.v47i1. 60592.
- Helal, James, Stephan, André, Crawford, Robert H., 2020. The influence of structural design methods on the embodied greenhouse gas emissions of structural systems for tall buildings. In: Structures, vol. 24, Elsevier, pp. 650–665. http://dx.doi.org/ 10.1016/j.istruc.2020.01.026.
- Hertwich, Edgar G., Ali, Saleem, Ciacci, Luca, et al., 2019. Material efficiency strategies to reducing greenhouse gas emissions associated with buildings, vehicles, and electronics-a review. Environ. Res. Lett. 14 (4), http://dx.doi.org/10.1088/1748-9326/ab0fe3.
- Höjer, Mattias, Wangel, Josefin, 2015. Smart sustainable cities: definition and challenges. In: ICT Innovations for Sustainability. Springer, pp. 333–349. http://dx.doi.org/10.1007/978-3-319-09228-7_20.
- Hosen, N., Nakamura, H., Hamzah, A., 2019. Traditional ecological knowledge and climate change adaptation: the sa'ban experience. J. Asian Behav. Stud. 4 (14), 63–77. http://dx.doi.org/10.21834/jabs.v4i14.339.
- Hossaini, Navid, Reza, Bahareh, Akhtar, Sharmin, et al., 2014. AHP based life cycle sustainability assessment (LCSA) framework: A case study of six storey wood frame and concrete frame buildings in vancouver. J. Environ. Plan. Manag. 58 (7), 1217–1241. http://dx.doi.org/10.1080/09640568.2014.920704.

- Kaewunruen, Sakdirat, Peng, Shijie, Phil-Ebosie, Olisa, 2020. Digital twin aided sustainability and vulnerability audit for subway stations. Sustainability 12 (19), 7873. http://dx.doi.org/10.3390/su12197873.
- Kamali, Mohammad, Hewage, Kasun, Sadiq, Rehan, 2019. Conventional versus modular construction methods: A comparative cradle-to-gate LCA for residential buildings. Energy Build. 204, 109479. http://dx.doi.org/10.1016/j.enbuild.2019.109479.
- Keoleian, Gregory A., Blanchard, Stephen, Reppe, Patricia, 2003. Life cycle energy, costs, and strategies for improving a single-family house. J. Ind. Ecol. 7 (2), 135–156. http://dx.doi.org/10.1162/108819800569726.
- Khan, S., Fernandez, G., 2019. A survey of digital twin technologies in civil engineering. Adv. Eng. Informatics 42, 100942. http://dx.doi.org/10.1016/j.aei.2019.100942.
- Kibert, Charles J., 2016. Sustainable Construction: Green Building Design and Delivery. John Wiley & Sons.
- Kineber, Ahmed Farouk, Singh, Atul Kumar, Fazeli, Abdulwahed, et al., 2023. Modelling the relationship between digital twins implementation barriers and sustainability pillars: Insights from building and construction sector. Sustain. Cities Soc. 99, 104930. http://dx.doi.org/10.1016/j.scs.2023.104930.
- Kiss, Benedek, Szalay, Zsuzsa, 2020. Modular approach to multi-objective environmental optimization of buildings. Autom. Constr. 111, 103044. http://dx.doi.org/10.1016/ i.autcon.2019.103044.
- Kumar, Sachin, Rodrigues, Joel J.P.C., Zymbler, Mikhail L., 2019. Internet of things is a revolutionary approach for future technology enhancement: A review. J. Big Data 6 (1), 1–21. http://dx.doi.org/10.1186/s40537-019-0268-2.
- Kylili, A., Fokaides, P.A., 2019. Construction materials for the urban environment: Environmental assessment of life cycle performance. In: Handbook of ecomaterials. Springer, Cham, Switzerland, http://dx.doi.org/10.1007/978-3-319-68255-6_133.
- Li, Z., Ye, J., Gao, B., et al., 2022. Digital and automatic design of free-form single-layer grid structures. Autom. Constr. 133, 104025. http://dx.doi.org/10.1016/j.autcon. 2021.104025.
- Liu, Zhenyuan, Hansen, Daniel Wilhelm, Chen, Ziyue, 2023a. Leveraging digital twins to support industrial symbiosis networks: A case study in the norwegian wood supply chain collaboration. Sustainability 15 (3), 2647. http://dx.doi.org/10.3390/ su15032647.
- Liu, Y., Liu, Z., Wang, H., 2023b. Leveraging BIM and IoT for smart facility management. Buildings 14 (3), 702. http://dx.doi.org/10.3390/buildings14030702.
- Lu, Weisheng, Tan, Tan, Xu, Jinying, Wang, Jing, Chen, Ke, Gao, Shang, Xue, Fan, 2021. Design for manufacture and assembly (DfMA) in construction: The old and the new. Archit. Eng. Des. Manag. 17 (1–2), 77–91.
- Madni, Azad M., Madni, Carla C., Lucero, Scott D., 2019. Leveraging digital twin technology in model-based systems engineering. Systems 7 (1), 7. http://dx.doi. org/10.3390/systems7010007.
- Morales, Michele Ferreira Dias, Reguly, Natalia, Kirchheim, Ana Paula, et al., 2020. Uncertainties related to the replacement stage in LCA of buildings: A case study of a structural masonry clay hollow brick wall. J. Clean. Prod. 251, 119649. http://dx.doi.org/10.1016/j.jclepro.2019.119649.
- Moser, Susanne C., Williams, Sarah J., Boesch, Donald F., 2015. Wicked challenges at land's end: Managing coastal vulnerability under climate change. Annu. Rev. Environ. Resour. 41, 51–78. http://dx.doi.org/10.1146/annurev-environ-021611-135158.
- Mukherjee, S., Chatterjee, K., Ghosh, S., 2022. Application of BIM in heritage buildings: A case study. J. Cult. Herit. Manag. Sustain. Dev. 12 (2), 134–150. http://dx.doi. org/10.1108/JCHMSD-06-2021-0105.
- Mulyasari, G., Trisusilo, A., Windirah, N., et al., 2023. Assessing perceptions and adaptation responses to climate change among small-scale fishery on the northern coastal of Bengkulu, Indonesia. Sci. World J. 2023, 1–15. http://dx.doi.org/10. 1155/2023/8770267.
- Mushi, V., Edward, M., 2021. Climate change adaptation practices for sustainable food production in Rombo district, Tanzania. J. Geogr. Assoc. Tanzan. 36 (2), http://dx.doi.org/10.56279/jgat.v36i2.151.
- Ofuoku, A., 2011. Rural farmers' perception of climate change in central agricultural zone of delta state, Nigeria. Indones. J. Agric. Sci. 12 (2), 63. http://dx.doi.org/ 10.21082/ijas.v12n2.2011.p63-69.
- Okoro, I., Okoro, P., Monday, S., et al., 2022. Climate change adaptation strategies: implication for behavioural change. Asian J. Environ. Ecol. 18 (4), 1–9. http: //dx.doi.org/10.9734/ajee/2022/v18i430322.
- Oramah, C., Olsen, O., 2021. Equity and justice in climate change adaptation: policy and practical implication in Nigeria. In: Justice and Climate Change. Springer, pp. 1767–1787. http://dx.doi.org/10.1007/978-3-030-45106-6_45.
- Padhiary, Mrutyunjay, Roy, Pankaj, Roy, Dipak, 2024. The future of urban connectivity. In: Advances in Electronic Government, Digital Divide, and Regional Development. IGI Global, pp. 33–66.
- Peterson, G., Leo, G., Hellmann, J., et al., 1997. Uncertainty, climate change, and adaptive management. Conserv. Ecol. 1 (2), http://dx.doi.org/10.5751/ES-00024-010204.
- Pittau, Francesco, Lumia, Gabriele, Heeren, Niko, et al., 2019. Retrofit as a carbon sink: The carbon storage potentials of the EU housing stock. J. Clean. Prod. 214, 365–376. http://dx.doi.org/10.1016/j.jclepro.2018.12.304.
- Płoszaj-Mazurek, Mateusz, Ryńska, Elżbieta, Grochulska-Salak, Magdalena, 2020. Methods to optimize carbon footprint of buildings in regenerative architectural design with the use of machine learning, convolutional neural network, and parametric design. Energies 13 (20), 5289. http://dx.doi.org/10.3390/en13205289.

- Shanbhag, Sejal Sanjay, Dixit, Manish Kumar, 2025. Integrating evolving energy economy scenarios in dynamic life cycle assessment of buildings. Environ. Impact Assess. Rev. 112, 107772. http://dx.doi.org/10.1016/j.eiar.2024.107772, URL https://www.sciencedirect.com/science/article/pii/S0195925524003597.
- Shrestha, Jagat Kumar, 2021. Assessment of energy demand and greenhouse gas emissions in low rise building systems: Case study of five building systems built after the Gorkha Earthquake in Nepal. J. Build. Eng. 34, 101831. http://dx.doi. org/10.1016/j.jobe.2020.101831.
- Sokolova, Marina V., Fernández-Caballero, Antonio, 2010. Environmental impact assessment by multi-agent systems. In: Agent and Multi-Agent Technology for Internet and Enterprise Systems. Springer, pp. 69–89.
- Soust-Verdaguer, Bernardette, Llatas, Carmen, García-Martínez, Antonio, 2017. Critical review of bim-based LCA method to buildings. Energy Build. 136, 110–120. http://dx.doi.org/10.1016/j.enbuild.2016.12.009.
- Spinelle, Laurent, Gerboles, Michel, Villani, Maria Gabriella, Aleixandre, Manuel, Bonavitacola, Fausto, 2015. Field calibration of a cluster of low-cost available sensors for air quality monitoring. Part A: Ozone and nitrogen dioxide. Sensors Actuators B 215, 249–257. http://dx.doi.org/10.1016/j.snb.2015.03.031, URL https://www.sciencedirect.com/science/article/pii/S092540051500355X.
- Swai, O., Mbwambo, J., Magayane, F., 2012. Gender and adaptation practices to the effects of climate change in Bahi and Kondoa districts Dodoma region, Tanzania. J. Sustain. Dev. 5 (12), 65. http://dx.doi.org/10.5539/jsd.v5n12p65.
- Tao, Fei, Zhang, He, Liu, Andy, et al., 2018. Digital twin-the simulation aspect. In: Digital Twin Driven Smart Manufacturing. Academic Press, pp. 59–74. http: //dx.doi.org/10.1016/B978-0-12-814435-9.00005-X.
- Tavana, Madjid, Izadikhah, Mohammad, Farzipoor Saen, Reza, et al., 2021. An integrated data envelopment analysis and life cycle assessment method for performance measurement in green construction management. Environ. Sci. Pollut. Res. 28 (1), 664–682. http://dx.doi.org/10.1007/s11356-020-10353-7.

- Thiombiano, B., Ouedraogo, D., Le, Q., et al., 2018. Role of climate change awareness in sustainable soil nutrient management by smallholder farms in Burkina Faso. Afr. J. Agric. Res. 13 (6), 294–301. http://dx.doi.org/10.5897/ajar2017.12939.
- Wastiels, L., Decuypere, R., 2019. Identification and comparison of LCA-BIM integration strategies. In: IOP Conference Series: Earth and Environmental Scienc, vol. 323, IOP Publishing, 012101. http://dx.doi.org/10.1088/1755-1315/323/1/012101.
- Xia, H., Liu, Z., Efremochkina, M., Liu, X., Lin, C., 2022. Study on city digital twin technologies for sustainable smart city design: A review and bibliometric analysis of geographic information system and building information modelling integration. Sustain. Cities Soc. 84, 104009. http://dx.doi.org/10.1016/j.scs.2022.104009.
- Xie, Y., Allen, C., Ali, S., 2019. Digital twin and big data towards smart manufacturing and industry 4.0: 360 degree comparison. IEEE Access 7, 3585–3593. http://dx. doi.org/10.1109/ACCESS.2018.2793265.
- Ylmén, Peter, Berlin, Johanna, Mjörnell, Kristina, et al., 2020. Managing choice uncertainties in life-cycle assessment as a decision-support tool for building design: A case study on building framework. Sustainability 12 (12), 5130. http://dx.doi. org/10.3390/su12125130.
- Zeng, Ruochen, Chini, Abdol, Ries, Robert, 2020. Innovative design for sustainability: integrating embodied impacts and costs during the early design phase. Eng. Constr. Archit. Manag. http://dx.doi.org/10.1108/ECAM-09-2019-0491.
- Zheng, Zibin, Xie, Shaoan, Dai, Hongning, et al., 2017. An overview of blockchain technology: Architecture, consensus, and future trends. In: 2017 IEEE International Congress on Big Data. BigData Congress, pp. 557–564. http://dx.doi.org/10.1109/ BigDataCongress.2017.85.
- Zook, Matthew A., Graham, Mark, 2007. Mapping DigiPlace: geocoded internet data and the representation of place. Environ. Plan. B: Plan. Des. 34 (3), 466–482. http://dx.doi.org/10.1068/b3311.